Author's response to reviews

Title: Predictors of Suboptimal CD4 Response among Women Achieving Virologic Suppression in a Randomized Antiretroviral Treatment Trial, Africa

Authors:

Aida Asmelash (aasmelash@bhp.org.bw)
Yu Zheng (ezheng@sdac.harvard.edu)
Kara Wools-Kaloustian (kwools@iupui.edu)
Douglas Shaffer (adx9@cdc.gov)
Fred Sawe (fsawe@hivresearch.org)
Anthony Ogwu (aogwu@bhp.org.bw)
Robert Salata (ras7@po.cwru.edu)
Judith Currier (jiscurrier@mednet.ucla.edu)
Michael Hughes (m Hughes@sdac.harvard.edu)
shahin Lockman (shahin.lockman@gmail.com)

Version: 3
Date: 25 October 2013

Author's response to reviews: see over
October 24, 2013
To BMC Reviewers:

The submission for “Predictors of Suboptimal CD4 Response among Women Achieving Virologic Suppression in a Randomized Antiretroviral Treatment Trial, Africa” was revised.

1) The abstract was restructured per the guidelines of BMC. It was divided to Background, Methods, Results, and Conclusion.

2) The Trials registration number was moved after the abstract.

3) A statement that research subjects signed informed consent was added at the methods section.

4) A list of all the Institutional review boards that reviewed the protocol and approved the protocol was added at the end of the methods section.

5) The tables and figures are separated from the body of the manuscript and will be uploaded separately.

6) The body of the manuscripts was edited again and some semantics corrected.

Sincerely,

Aida Asmelash MD, MPH
Corresponding Author.